Author's response to reviews

Title: Male Gender is Independently Associated with Pulmonary Tuberculosis among sputum and non-sputum producers Suspects in Southwestern Uganda

Authors:

yap JB Boum II (yap.boum@epicentre.msf.org)
Daniel Atwine (daniel.atwine@epicentre.msf.org)
Patrick Orikiriza (patrick.orikiriza@epicentre.msf.org)
justus assimwe (asiimwejpk@yahoo.co.uk)
Anne-Laure Page (anne-laure.page@epicentre.msf.org)
Juliet Mwanga Amumpaire (juliet.mwanga@epicentre.msf.org)
Maryline Bonnet (maryline.bonnet@epicentre.msf.org)

Version: 2
Date: 20 June 2014

Author's response to reviews:

The revised version include the following sections: Competing interests, Authors contribution. the legends for the figures have been added as well as the authors e-mail adresses.